\begin{table}%t1
\caption{Photometry of HR~8799.}
\label{tab:photometry}
%%\centerline

{
%\tiny
\begin{tabular}{lllr}
\hline
\hline
Photometric band  & Flux or magnitude     &
$F_{\rm qual}$ (IRAS) & Ref.  \\
\hline
  & [mag]     &   &   \\
\hline
$B$ & 6.090 $\pm$ 0.300 &   & (1) \\
$B$ & 6.196     &   & (2) \\
$B$ & 6.210 $\pm$ 0.010 &   & (3) \\
$B$ & 6.214 $\pm$ 0.009 &   & (4) \\
$V$ & 5.960 $\pm$ 0.010 &   & (4) \\
$V$ & 5.959     &   & (2) \\
$V$ & 5.960 $\pm$ 0.010 &   & (3) \\
$R$ & 5.810 $\pm$ 0.300 &   & (1) \\
$I$ & 5.690 $\pm$ 0.300 &   & (1) \\
$J$ & 5.383 $\pm$ 0.027 &   & (5) \\
$H$ & 5.280 $\pm$ 0.018 &   & (5) \\
$K_{\rm s}$ & 5.240 $\pm$ 0.018 &   & (5) \\
\hline
  & [Jy]      &   &     \\
\hline
IRAS PSC 12 $\mu$m  & 0.267 $\pm$ 0.034 & 3 & (6) \\
IRAS PSC 25 $\mu$m  & 0.246 $\pm$ 0.000 & 1 & (6) \\
IRAS PSC 60 $\mu$m  & 0.307 $\pm$ 0.061 & 2 & (6) \\
IRAS PSC 100 $\mu$m & 2.376 $\pm$ 0.000 & 1 & (6) \\
IRAS FSC 12 $\mu$m  & 0.278 $\pm$ 0.036 & 3 & (7) \\
IRAS FSC 25 $\mu$m  & 0.174 $\pm$ 0.075 & 1 & (7) \\
IRAS FSC 60 $\mu$m  & 0.311 $\pm$ 0.062 & 3 & (7) \\
IRAS FSC 100 $\mu$m & 3.202 $\pm$ 0.977 & 1 & (7) \\
ISO 60 $\mu$m         & 0.412 $\pm$ 0.021 &   & (8) \\
ISO 90 $\mu$m         & 0.585 $\pm$ 0.041 &   & (8) \\
\hline
  & [mJy]     &   &           \\
\hline
JCMT 850 $\mu$m & 10.3  $\pm$ 1.8   &   & (9) \\
JCMT 1100 $\mu$m  & $<$33     &     & (10)  \\
IRAM 1200 $\mu$m  & 4.8 $\pm$ 2.7 &     & (10)  \\
\hline
\end{tabular}}
\medskip
\par
IRAS fluxes are color corrected as described in the IRAS Explanatory Supplement \citep{beichman-et-al-1988}. \\
References:
(1) The USNO-B1.0 Catalog \citep{monet-et-al-2003};
(2) NOMAD Catalog \citep{zacharias-et-al-2004}, from Tycho-2 Catalog
\citep{hog-et-al-2000};
(3) The Guide Star Catalog, Version 2.3.2
    \citep{lasker-et-al-2008};
(4) The Hipparcos and Tycho Catalogues \citep{perryman-et-al-1997};
(5) 2MASS All-Sky Catalog of Point Sources \citep{skrutskie-et-al-2006};
(6) IRAS catalogue of Point Sources, Version 2.0
\citep{helou-walker-1988};
(7) IRAS Faint Source Catalog, $|b|$ $>$ 10, Version 2.0
\citep{moshir-et-al-1990};
(8) \citep{moor-et-al-2006}
(9) \citep{williams-andrews-2006}
(10) \citep{sylvester-et-al-1996}
\end{table}